Author's response to reviews

Title: Monitoring the referral system through benchmarking in rural Niger: an evaluation of the functional relation between health centres and the district hospital

Authors:

Paul Bossyns (paul.bossyns@pandora.be)
Ranaou Abache (Aranaoudf@yahoo.fr)
Mahaman S Abdoulaye (Abdoulayemsani@yahoo.fr)
Hamidou Miye (Hasmihamidou@hotmail.com)
Anne-Marie Depoorter (Anne-Marie.Depoorter@vub.ac.be)
Wim Van Lerberghe (vanlerberghe@who.int)

Version: 5 Date: 9 April 2006

Author's response to reviews:

Title and abstract were copied from manuscript as to correspond exactly

Author list: Names were adjusted

Authors contributions: removed from title page

Keywords section: removed

Material and methods: changed into methods

References: "et al" not used any longer

Figure legend: placed just after references

Figure in word: cropped